Author’s response to reviews

Title: Relationship between arsenic skin lesion and the age of natural menopause

Authors:

Fakir Md Yunus (dryunus155@gmail.com)
Musarrat Jabeen Rahman (jabeen@icddrb.org)
Md. Zahidul Alam (mzalam@uta.edu)
Samar Kumar Hore (samar@icddrb.org)
Mahfuzar Rahman (mrahman1@uchicago.edu)

Version: 6 Date: 30 March 2014

Author’s response to reviews: see over
Subject: Response to the Editor’s comments on manuscript.

Dear Sir,

Many thanks for your valuable comments on the manuscript ‘Relationship between arsenic skin lesion and the age of natural menopause’. As per your recommendation, we invited Maria A May from BRAC Social Innovation Lab to copyedit our manuscript in order to improve the style of written English, and we also acknowledged her. Therefore, we are looking forward to publish this original research article in your reputed journal soon. We humbly request for your indulgence.

Finally, we appreciate your time and hard work on our manuscript.

Your sincerely-

Corresponding author on behalf of all authors

Fakir Md. Yunus MBBS. MPH
Senior Research Associate
BRAC Research and Evaluation Division.
BRAC Centre, 75 Mohakhali, Dhaka-1212
Bangladesh.
dryunus155@gmail.com
yunus.fm@brac.net